

G OPEN ACCESS

Citation: Papandrianos N, Papageorgiou E, Anagnostis A, Papageorgiou K (2020) Bone metastasis classification using whole body images from prostate cancer patients based on convolutional neural networks application. PLoS ONE 15(8): e0237213. https://doi.org/10.1371/ journal.pone.0237213

Editor: Jeonghwan Gwak, Korea National University of Transportation, REPUBLIC OF KOREA

Received: November 16, 2019

Accepted: July 22, 2020

Published: August 14, 2020

Copyright: © 2020 Papandrianos et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data (which are anonymized bone scintigraphy images without any further information) are available only after request for research purposes. Data availability: The dataset from Diagnostic Medical Center "Diagnostico-latriki A.E." was used under the license of the Board Committee Director of the Diagnostic Medical Center "Diagnostico-latriki A. E." Dr. Vasilios Parafestas for the current study, and is not publicly available. The dataset may be available only under request from the Director of **RESEARCH ARTICLE**

Bone metastasis classification using whole body images from prostate cancer patients based on convolutional neural networks application

Nikolaos Papandrianos¹*, Elpiniki Papageorgiou^{2,3}, Athanasios Anagnostis^{3,4}, Konstantinos Papageorgiou⁴

1 General Department, University of Thessaly, Lamia, Greece, 2 Faculty of Technology, Dept. of Energy Systems, University of Thessaly, Geopolis Campus, Larisa, Greece, 3 Institute for Bio-economy and Agritechnology, Center for Research and Technology Hellas, Greece, 4 Department of Computer Science and Telecommunications, University of Thessaly, Lamia, Greece

* npapandrianos@uth.gr

Abstract

Bone metastasis is one of the most frequent diseases in prostate cancer; scintigraphy imaging is particularly important for the clinical diagnosis of bone metastasis. Up to date, minimal research has been conducted regarding the application of machine learning with emphasis on modern efficient convolutional neural networks (CNNs) algorithms, for the diagnosis of prostate cancer metastasis from bone scintigraphy images. The advantageous and outstanding capabilities of deep learning, machine learning's groundbreaking technological advancement, have not yet been fully investigated regarding their application in computeraided diagnosis systems in the field of medical image analysis, such as the problem of bone metastasis classification in whole-body scans. In particular, CNNs are gaining great attention due to their ability to recognize complex visual patterns, in the same way as human perception operates. Considering all these new enhancements in the field of deep learning, a set of simpler, faster and more accurate CNN architectures, designed for classification of metastatic prostate cancer in bones, is explored. This research study has a two-fold goal: to create and also demonstrate a set of simple but robust CNN models for automatic classification of whole-body scans in two categories, malignant (bone metastasis) or healthy, using solely the scans at the input level. Through a meticulous exploration of CNN hyper-parameter selection and fine-tuning, the best architecture is selected with respect to classification accuracy. Thus a CNN model with improved classification capabilities for bone metastasis diagnosis is produced, using bone scans from prostate cancer patients. The achieved classification testing accuracy is 97.38%, whereas the average sensitivity is approximately 95.8%. Finally, the best-performing CNN method is compared to other popular and wellknown CNN architectures used for medical imaging, like VGG16, ResNet50, GoogleNet and MobileNet. The classification results show that the proposed CNN-based approach outperforms the popular CNN methods in nuclear medicine for metastatic prostate cancer diagnosis in bones.

the Diagnostic Center (vparafestas@yahoo.gr) and the relevant Doctor of Nuclear Medicine Dr. Nikolaos Papandrianos, (npapandrianos@uth.gr; nikpapandr@gmail.com).

Funding: The author(s) received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

1. Introduction

Most common tumors, such as those of the breast, lung and prostate, frequently metastasize to the bone tissue, and so, the skeleton seems to be a site with the most significant tumor burden in cancer patients with advanced disease. Statistical analysis results have shown that 65% of all bone metastases originate from the breast in women and from the prostate in men. The remaining 35% emanates from thyroid, lung and kidney cancers [1]. In the case of metastatic prostate cancer, diagnosis has a significant impact on the quality of patient's life [2]. In most men, the metastatic prostate cancer mainly sites on the bones of the axial skeleton, causing severe lesions that can cause pain, debility and/or functional impairment [3]. As this type of cancer has great avidity for bone and could cause painful and untreatable effects, an early diagnosis is crucial for the patient. Reviews on clinical evidences and diagnostic assessments of bone metastases in men with prostate cancer can be found in [4].

The implementation of a properly selected diagnostic imaging can reveal the number of metastatic foci in the skeletal system [1, 5]. Rapid diagnosis of bone metastases can be achieved using modern imaging techniques such as scintigraphy, Positron Emission Tomography (PET) and whole-body Magnetic Resonance Imaging (MRI).

The primary imaging method in the diagnosis of metastases, that offers the highest sensitivity among all imaging methods (95%), is Bone Scintigraphy (BS) [6–8]. Through the depiction of the entire skeleton in one medical examination, nuclear doctor is able to detect bone abnormalities in areas where intensive radionuclide activity is present. However, low specificity seems to be the main drawback of this method, as it cannot tell whether the causes of bone turnovers are different than those of metastatic origin (leukaemia, healing fracture, etc.). At the same time, PET has been recognized as an efficient method for detecting cancer cells, based on recent technological advancements in medical imaging. PET and Computed Tomography (CT) combination can produce high-resolution images [9–11].

Although PET and PET/CT are the most efficient screening techniques for bone metastasis, BS remains the most common imaging procedure in nuclear medicine [9, 12], [13]. As reported in European Association of Nuclear Medicine (EANM) guidelines [13], BS is particularly important for clinical diagnosis of metastatic cancer, both in men and women. At present, when other imaging or examination methods are unable to provide a reliable diagnosis, BS imaging becomes the proper modality for making a final diagnosis of bone metastasis [14].

To address the considerable problem of bone metastasis diagnosis, artificial intelligent methods for medical image analysis, implemented with deep learning algorithms, have been adequately investigated. In this direction, a recent survey reveals the entire penetration of deep learning techniques into the field of medical image analysis; detection, segmentation, classification, retrieval, image generation and enhancement, registration and successful application of deep learning to medical imaging tasks are thoroughly examined [15–18].

Implementation of deep learning in medical imaging is mainly conducted by Convolutional Neural Networks (CNNs) [16, 19, 20], a relatively new and powerful way to learn useful representations of images and other structured data. Before the application of CNNs, these features typically had to be created by less powerful machine learning models or even hand-crafted. With the introduction of CNNs, such features could be learned directly from the provided data, since they include certain preferences in their structure that make them powerful deep learning models for image analysis [15, 20, 21]. Typical CNNs have a similar structure with Artificial Neural Networks (ANN) and consist of one or more filters (i.e., convolutional layers), followed by aggregation/pooling layers in order to extract features for classification tasks [22]. Gradient descent and backpropagation are both used as learning algorithms, the same way they are used in a standard ANN. Their main difference lies in the fact that CNNs have

layers of convolutions along with pooling layers in the beginning of their architecture. The final outputs are computed via fully connected layers, located at the end of the network architecture [18].

In recent years, CNNs have gained wider recognition in medical image analysis domain, as well as in vision systems [15–18, 20]. Due to their enormous popularity, several applications of CNNs were investigated in the field of medical image analysis. Two recent review studies, [15] and [18], gather all the important and most interesting applications of deep learning.

The application of CNNs in medical imaging ranges from plain radiograph, CT, MRI and microscopy images to clinical photos (dermatology), capsule endoscopy and visual recognition [23–29]. In addition, a CNN was investigated in [30] that regards automatic detection of tuber-culosis on chest radiographs, while in [31], a brain tumor segmentation in magnetic resonance images was made possible with the use of a CNN. More examples of CNNs' successful application in medical domain include automated cardiac diagnosis [32], detection of lesions and prediction of treatment response by PET [33, 34], as well as dynamic contrast agent-enhanced computed tomography, where CNN showed high diagnostic performance in the differentiation of liver masses [35]. Furthermore, CNNs have shown outstanding performance in radiology and molecular imaging [25].

Some models with major impact in the context of deep learning and medical image processing were introduced in several research articles: the U-net model for biomedical data semantic segmentation [36], the GoogLeNet model introducing the inception module [37], the ResNet model introducing the residual learning building block for extremely deep convolutional networks [38] and also, Deeplab, which deals with the inclusion of many convolutional layers atrous for semantic segmentation of images in deep convolutional neural networks [39].

In medical image analysis, the most widely used CNN methods are the following: (i) Alex-Net (2012) [40]: This network has a quite deep architecture, similar to GoogLeNet, by Yann LeCun et. al [41], incorporates more filters per layer and includes stacked convolutional layers. It attaches ReLU activations after every convolutional and fully-connected layer. (ii) ZFNet (2013) [42]: Being a rather slight modification of AlexNet, this network won the 2013 ILSVRC competition. (iii) VGGNet16 (2014) [43, 44]: It consists of 16 convolutional layers, having a very uniform architecture, similar to AlexNet. (iv) GoogleNet [37]: It is a convolutional neural network with a standard stacked convolutional layer, having one or more fully connected layers, called inception modules, able to extract various levels of features on the same time. (v) ResNet (2015) [38]: It is another efficient CNN architecture that introduced the "identity shortcut connection", to solve the notorious problem of the vanishing gradients of the deep networks. (vi) DenseNet (2017) [45]: Being another important CNN architecture, DenseNet offers the main advantage of alleviating the gradient vanishment problem with the direct connection of all the layers.

1.1 Related work in nuclear medical imaging for metastatic prostate cancer diagnosis in bones

Reviewing the relevant literature for diagnosis of bone metastasis using bone scintigraphy scans, the authors notice that only a couple of previous works have been adequately conducted for metastatic prostate cancer classification using CNNs, while the others are devoted to ANNs and their application in Computer-Aided Diagnosis (CAD). These works have investigated the use of Bone Scan Index (BSI), which was introduced to assess the bone scanning process and estimate the extent of bone metastasis [46, 47]. Specifically, it serves as a clinical, quantitative and reproducible parameter that can measure metastatic prostate cancer bone involvement [47]. The software developed for the BSI-based ANN approach, was EXINI bone (EXINI

Diagnostics AB, Lund, Sweden) and afterwards, a revised version of this software, called BONENAVI (FUJIFILM Toyama Chemical, Co. Ltd, Tokyo, Japan), was engineered using a large number of Japanese multicenter training databases [48].

The first of the reported studies was devoted to the development of a classification algorithm based on CNNs for bone scintigraphy image analysis [49]. It was carried out as a master thesis in Lund University and is focused mainly on classification problems, without considering any identification and segmentation tasks. The used dataset was provided by Exini Diagnostics AB, in the form of image patches of already found hotspots. The process in which hotspots were segmented, cropped and collected from bone scans, was implemented using a software developed at Exini. BSI was calculated for whole-body bone scans, by segmenting the entire skeleton from the background in both the anterior and posterior views. Due to time frame restrictions, only the hotspots found in the spine have been used to train the CNN, since they were considered to be the easiest to classify. A shape model based on a mean shape of several normal whole-body scans, was fitted to the skeleton, using an image analysis algorithm, called Morphon registration. The outcomes of the aforementioned thesis [49] have shown that the calculated accuracy of the validation set was 0.875, whereas the calculated accuracy of the testing set was 0.89.

The second study explored CNNs for classification of prostate cancer metastases using bone scan images [50]. The tasks of this master project appeared to have a significant potential on classifying bone scan images obtained by Exini Diagnostics AB too, including BSI. The two tasks were defined as: i) classifying anterior / posterior pose and ii) classifying metastatic / non-metastatic hotspots. The outcome of this study is that the trained models produce highly accurate results in both tasks and they outperform other methods for all tested body regions in the case of metastatic / non-metastatic hotspots classification. The evaluation indicator of the area under Receiver Operating Characteristic (ROC) score was equal to 0.9739, which is significantly higher than the respective ROC of 0.9352, obtained by methods reported in the literature for the same test set.

The remaining research that concerns the same imaging modality (BS), is devoted to the introduction of CAD systems with the use of ANN and other Machine Learning (ML) methods for bone metastasis detection in bone scintigraphy images. Sadik et al. were the first to develop an automated CAD system as a clinical quality assurance tool, for the interpretation of bone scans [51–53]. This bone scan CAD software was trained to interpret bone scans using training databases that consist of bone scans from European patients who have the desired image interpretation, metastatic disease or not. The results showed a sensitivity of 90% at a specificity of 74%. These works result in certain outcomes that refer to the development of a totally automated computer-assisted diagnosis system that can identify metastases after examining bone scans, applying multi-layer perceptron ANN techniques, involving a small database of whole-body bone scans (135 patients). The highest sensitivity that was achieved from all the studies and accomplished during this thesis, was approximately 89% [54].

Horikoshi et al. compared the diagnostic accuracy of two CAD systems, one based on a European and another on a Japanese training database, in a group of bone scans from Japanese patients [48]. The Japanese CAD software showed a higher specificity and accuracy compared to the European. Comparing the sensitivities, the Japanese CAD software achieved 90%, whereas the European CAD software reached 83% [48].

In another study conducted by Tokuda et al., the diagnostic capability of a completely automated CAD system, which detects metastases in the images of bone scans by focusing on two different patterns, was investigated [55]. The first pattern was devoted to the detection of metastases per region; the second one detects metastases per patient. The investigated system was called "BONENAVI version 1". The produced results have shown that the new CAD system is able to decrease the number of false positive findings, which depends on the primary lesion of cancer.

In 2016, Aslantas et al. proposed "CADBOSS" as a fully automated diagnosis system for bone metastases detections, using whole-body images [56]. The proposed CAD system combines an active contour segmentation algorithm for hotspots detection, an advanced method of image gridding to extract certain characteristics of metastatic regions, as well as an ANN classifier for identifying possible metastases. The calculated accuracy, sensitivity, and specificity of CADBOSS were 92.30%, 94%, and 86.67%, respectively, outperforming other state of the art CAD systems.

Additionally, ML methods have been exploited and applied in CAD systems for bone metastasis detection in bone scintigraphy images. A parallelepiped classification method was specifically deployed in [57] to assist physicians in bone metastases detection of cancer. Decision Trees (DT) and Support Vector Machines (SVM) were exploited for predicting skeletal-related events in cancer patients with bone metastases, achieving higher accuracies with a smaller number of variables than the number of variables used in Linear Regression (LR). ML techniques can be also used to build accurate models to predict skeletal-related events in cancer patients with bone metastasis, providing an overall classification accuracy of 87.58% \pm 2.25% [58].

As far as PET and PET/CT imaging techniques in nuclear medicine are concerned, there are some recent and prominent studies that apply the advantageous features of CNNs. In [59], deep learning has been applied for classification of benign and malignant bone lesions in [F-18]NaF PET/CT images. The authors in this work followed the VGG19 architecture for their network by employing 16 3×3 convolutional layers, followed by 2 fully connected layers and a softmax layer as final activation. The ImageNet database of natural images was further used to pre-train the network's weights. In this way, the network first is trained on general image features and later is tuned using the lesion images and the physician's scores. Taking a closer look at the results, it can be concluded that network's performance was improved when it was trained to differentiate between definitely benign (score = 1) and definitely malignant (score = 5) lesions. The values of prediction metrics were 0.88, 0.90, 0.85 and 0.90, concerning the accuracy, sensitivity, specificity and positive predictive value, respectively.

Also, a CNN-based system was examined in a recent retrospective study, which included 3485 sequential patients who underwent whole-body FDG PET-CT [60]. The main purpose of the study was to detect malignant findings in FDG PET-CT examinations, while a neural network model, equivalent to ResNet24, was built. Additionally, Grad-CAM was employed to identify the part of the image on which the neural network used the largest information. The findings of the study showed that Grad-CAM reasonably highlighted the area of malignant uptake, allowing physicians to make a diagnosis. The same research team recently (2019) developed a CNN-based system that predicts the location of malignant uptake and further evaluated predictions accuracy [61]. A network model with configuration equivalent to ResNet24, was used to classify whole-body FDG PET images.

In the research work [62], a simple CNN-based system that predicts patient sex from FDG PET-CT images, was proposed. Specifically, 6462 consecutive patients have participated in the study and underwent whole-body FDG PET-CT. The CNN system was used for classifying these patients by sex. Another CNN-based diagnosis system for whole-body FDG PET-CT was developed in [63], that predicts whether physician's further diagnosis is required or not. A thorough analysis of the results shows that the accuracy considering images of patients presenting malignant uptake and images of equivocal was 93.2±3.9% and 87.8±5.3%, respectively.

The task of segmentation with the use of deep learning models in skeletal scintigraphy images, has been discussed in more research studies. For example, in [64], the authors followed

different approaches to convert convolutional neural networks, designed for classification tasks, into powerful pixel-wise predictors. Moreover, in [65], a deep-learning based segmentation method was developed using prostate-specific membrane antigen (PSMA) PET images and showed significant promise towards automated delineation and quantification of prostate cancer lesions. However, this research domain that involves bone scintigraphy segmentation with the inclusion of advanced deep neural networks has not been yet well established.

Although bone scintigraphy is extremely important for the diagnosis of metastatic cancer, it is clear that a small number of research works have been carried out, which presented only some preliminary results, while the advantageous and outstanding capabilities of CNNs have not been fully investigated. Reviewing the relevant literature, it appears that CNNs have not been sufficiently applied for the diagnosis of prostate cancer metastasis from whole body images.

1.2 Aim and contribution of this research work

CNN is an efficient deep learning network architecture that has recently found great applicability in the medical domain. It has shown excellent performance on medical image applications, including bone scintigraphy and nuclear medical imaging, and can offer a positive impact on diagnosis tasks.

Nowadays, the main challenge in bone scintigraphy, as being one of the most sensitive imaging methods in nuclear medicine, is to build an algorithm that automatically identifies whether a patient is suffering from bone metastasis or not, based on patient's whole body scans. It is of utmost importance that the algorithm needs to be extremely accurate due to the fact that patients' lives could be at stake. Deep learning algorithms whose potential lies in the fact that they can improve the accuracy of cancer screening, have been recently investigated for nuclear medical imaging analysis. Recent studies in BS and PET have shown that a deep learning-based system can perform as well as nuclear physicians do in standalone mode and improve physicians' performance in support mode. Even though BS is extremely important for the diagnosis of metastatic cancer, there is currently no research paper regarding the diagnosis of prostate cancer metastasis from whole body scan images that applies robust and more accurate deep CNNs.

This research study investigates the application of a deep learning CNN to classify bone metastasis using whole body images of men who were initially diagnosed with prostate cancer. The proposed method employs different CNN-based architectures with data normalization, data augmentation and shuffling in the preprocessing phase. In the training phase, the back-propagation technique has been used for updating the weights as part of the optimization process. Finally, the network architecture is fine-tuned and the configuration that offers the best performance is selected to train the CNN model.

The scope of the current work entails the following two main components: First, this study introduces the CNN method into the diagnosis of bone metastasis disease, based on whole body images. In the second phase, the paper deals with the improvement of the existing CNN method in terms of both network architecture and hyperparameter optimization. The deployed process seemingly improves the diagnostic effect of the deep learning method, making it more efficient compared to other benchmark and well-known CNN methods, such as ResNet50, VGG16, GoogleNET, Xception and MobileNet [66, 67].

The innovations and contributions of this paper are well summarized as follows:

• The development and demonstration of a simple, fast, robust CNN-based classification tool for the identification of bone metastasis in prostate cancer patients from whole-body scans.

- The rigorous CNN hyper-parameter exploration for determining the most appropriate architecture for an enhanced classification performance.
- A comparative experimental analysis, utilizing popular image classification CNN architectures, like ResNet50, VGG16, GoogleNET, Xception and MobileNet.

This paper is structured in the following fashion: Section 2 presents the material and methods related to this research study. Section 3 presents the proposed solution based on CNN method for bone metastasis diagnosis for prostate cancer patients using whole body images. Section 4 shows the results of exploration analysis of CNNs in Red, Green and Blue (RGB) mode, for different parameters and configurations, thus providing the best CNN model for this case study. Furthermore, all the performed experiments with representative results are gathered in section 4, whereas section 5 provides a thorough discussion on analysis of results. Section 6 concludes the paper and outlines future steps.

2. Materials and methods

2.1 Patients and images

A retrospective review of 970 consecutive, whole body scintigraphy images from 817 different male patients who visited Nuclear Medicine Department of Diagnostic Medical Center "Diagnostico A.E." in Larisa, Greece, from June 2013 to June 2017, was performed. The selection criterion was prostate cancer patients who had undergone whole-body scintigraphy, because of suspected bone metastatic disease.

Due to the fact that whole body scan images contain some artifacts and other non-related to bone uptake, such as urine contamination and medical accessories (i.e. urinary catheters) [68], as well as the frequent visible site of radiopharmaceutical injection [69], a preprocessing approach was accomplished to remove these artifacts and non-osseous uptake from the original images. This preprocessing method was accomplished by a nuclear medicine physician before the use of the dataset in the proposed classification approach.

The initial dataset of 970 images contained not only bone metastasis presence and absence patient's cases suffering from prostate cancer, but also degenerative lesions [70]. Due to this fact, as well as aiming to cope with a two-class classification problem in this study, a pre-selection process, concerning images of healthy and malignant patients, was accomplished. In specific, 586 out of 970 consecutive whole-body scintigraphy images of men from 507 different patients were selected and diagnosed accordingly by a nuclear medicine specialist, with 15 years of experience in bone scan interpretation. Out of 586 bone scan images, 368 bone scans concern male patients with bone metastasis and 218 male patients without bone metastasis. A nuclear medicine physician classified all the patient cases into 2 categories: 1) metastasis absent and 2) metastasis present, which was used as a gold standard (see Fig 1). The metastatic images were confirmed by further examinations performed by CT/MRI.

2.2 Whole-body scintigraphy (Bone scans)

A Siemens gamma camera Symbia S series SPECT System (by dedicated workstation and software Syngo VE32B) with two heads and low energy high-resolution collimators was used for patients scanning. The speed of scanning was 12 cm/min with no pixel zooming. Two types of radionuclide were used for bone scintigraphy: 99m-Tc-HDP (TechneScan®) and 99-Tc-MDP (PoltechMDP 5mg). Whole body scintigraphy was acquired approximately 3 hours after intravenous injection of 600–740 MBq of radiopharmaceutical agent, depending on the patient body type. The common intravenous injection was 670 MBq of radiopharmaceutical agent.



Fig 1. Image samples in the dataset (Label: (a) Metastasis is present or (b) absent). https://doi.org/10.1371/journal.pone.0237213.g001

In total, 586 planar bone scan images from patients with known prostate cancer were reviewed, retrospectively. The whole-body field was used to record anterior and posterior views digitally with resolution 1024×256 pixels. Images represent counts of detected gamma decays in each spatial unit with 16-bit grayscale depth.

The image data acquired were originally in DICOM files, a commonly used protocol for storage and communication in hospitals. These image data were extracted from these DICOM files to create new images in JPEG-format, instead. A novel dataset of bone scintigraphy images containing men patients suffering from prostate cancer with metastasis present and metastasis absent (two distinct classes of healthy and malignant cases), was prepared for experimentation. This dataset consisting of whole-body scans is available for research use after request.

This study was approved by the Board Committee Director of the Diagnostic Medical Center "Diagnostico-Iatriki A.E." and the requirement to obtain informed consent was waived by the Director of the Diagnostic Center due to its retrospective nature. All procedures in this study were in accordance with the Declaration of Helsinki.

2.3 Methodology

The problem of classifying bone metastasis images is a complex procedure and so, effective machine learning methods need to be exploited to cope with this diagnosis task. Deep learning methods such as CNNs are applied in order to train a classifier to distinguish images of prostate cancer patients with bone metastasis, and metastasis absent on healthy patients. The effective CNN method for bone metastasis classification proposed in this paper, includes three processing steps: data pre-processing for the collected scan data normalization, a training phase for CNN learning and validation, and testing which includes the evaluation of the classification results, as illustrated in Fig 2. The proposed methodology is thoroughly presented in the following sections.

2.3.1 Data pre-processing. *Step 1: Load images to RGB.* The original images were saved in RGB mode. All images are stored in a respective folder before loaded into the computer memory for CNN training. In each image, a suitable prefix was defined according to the patient's category, for example "malignant_" and "healthy_". Next, a small script was set up in order to



https://doi.org/10.1371/journal.pone.0237213.g002

assign a numerical value as label to each image, according to its prefix. In our case, the value '0' was assigned for "malignant_" prefixes, whereas the value '1' for "healthy_" prefixes.

Step 2: *Data normalization*. It is common to follow a normalization process (feature scaling) in most machine learning algorithms [70]. The min-max normalization processnormalizes the dataset values within the (0,1) to ensure that all feature data are in the same scale for training and testing. The data normalization process also assists the convergence of the backpropagation algorithm.

Step 3: *Data shuffle*. To avoid or eliminate unbiased sampling in machine learning, an appropriate shuffling method is needed to be defined. More specifically, a random-number generator is used to reorder the images. An image sample, chosen randomly, is meant to be an impartial representation of the total images. An unbiased random sample is important for machine learning to provide reliable conclusions. In this study, Python's random.shuffle method is used for dataset shuffling.

Step 4: *Data augmentation*. Data augmentation is used as a method to artificially increase the diversity of training data by a large margin, by manipulating the existing data instead of creating new. Data augmentation techniques such as cropping, padding, and horizontal flipping are commonly used to train large neural networks [71]. In this research, the number of images used for learning processes was followed by an image augmentation processing such as rotation, enlargement/reduction, range and flip. Note that the original images used for the test were not subjected to such an augmentation process.

Step 5: Data split. The dataset was split into three sections, a training portion, a validation portion that would allow the training process to improve, and a testing (hold-out) portion, which is part of the dataset that is completely hidden from the training process. The first data split takes place by removing 15% of the total dataset and saving for later use as testing. The remaining 85% of the dataset is then split again into an 80/20 ratio, where the small portion is used as validation set. The validation set is used during the training process in order to help the algorithm update its weights appropriately. Thus, it improves its performance and avoids overfitting. The testing dataset is used to verify if the unknown to the model data have been classified correctly

2.3.2 Convolutional neural networks–CNNs (Training and validation). To define a proper architecture of a Convolutional Neural Network (CNN) for image classification, an exploration process is needed. In the experimentation and trial phase, the most important parameters that can lead to effective network architecture were explored; the number of convolutional layers, the number of pooling layers, the number of nodes in the dense layers, the dropout rate and the batch size. In <u>S1 Appendix</u>, the main aspects of CNN implementation are provided. A typical CNN architecture is illustrated in Fig.3.

Next, a number of functions that characterize the CNN architecture is being defined through bibliographic research. The main functions include the activation function, which is the function that defines the output of the layer and the loss function, which is the function



Fig 3. Architecture of a convolutional neural network [72].

that is used for the optimization of the network's weights. Further explanation is provided in section 2.4.

The training process takes place when the algorithm tries to create a function that describes the desired relation, based on the training data. It then makes predictions based on this function and moves to the validation step. The validation dataset is used for the validation process in which, the main aim is the error minimization, that reveals the prediction efficiency of the proposed method. The minimization of the training loss and the validation loss are also desired outcomes for the validation phase to further demonstrate the model performance.

2.3.3 Testing and evaluation. The testing process for the CNN model is accomplished by using the testing data, which are completely unknown to the model. The classifier makes predictions on each image class and finally compares the calculated predicted class with the true class. After a classifier is trained, it can predict the class of a new entry from the testing set. Depending on the prediction and the actual class it belongs, this prediction can be true positive (TP) or true negative (TN), if it is classified correctly, or false positive (FP) or false negative (FN) if it is misclassified. Next, for classifier evaluation, some well-known and popular performance metrics, such as testing accuracy, precision, recall, sensitivity, specificity and F1-score of the model are computed [16, 19, 73] (see S2 Appendix). An error/confusion matrix is also employed to further evaluate the performance of the model. It consists of rows and columns, which show the correctly and wrongly classified images, in each column. Calculating the confusion matrix can give us a better view of the proposed classification model.

3. Proposed CNN architecture

In this research study, a CNN architecture is proposed to precisely identify bone metastasis from whole-body scans of men suffering from prostate cancer. The developed CNN will prove its capability to provide high accuracy with a simple and fast architecture for whole-body image classification. Through an extensive CNN exploration process, we conducted experiments with different values for our parameters, like pixels, epochs, drop rate, batch size, number of nodes and layers [25]. In common classic feature extraction techniques, a manual feature selection was required to extract and utilize the appropriate feature. Convolutional Neural Networks (CNN), resembling in type Artificial Neural Networks (ANN), can perform feature extraction techniques automatically by applying multiple filters on the input images and then, through an advanced learning process, they select those that have the highest impact on the images classification.

In this research study, a deep-layer network with 3 convolutional—pooling layers, 1 dense layer followed by a dropout layer, as well as a final output layer with one node, is built. The dedicated problem is binary classification. The suggested network architecture along with the layers shapes and the number of trainable parameters are given in Table 1.

The images enter the network at various pixels dimensions, starting from 100x100 pixels to 300x300 pixels. Following the structure of the CNN, the first (input) convolutional layer consists of 3 filters (kernels) of size 3x3, always followed by a max-pooling layer of size 2x2 and a dropout layer with 0.7 as dropout rate. Each next convolutional layer doubles the numbers of filters (8), just like the following max-pooling layers. Next, a flattening operation transforms the 2-dimensional matrices to 1-dimensional arrays, in order to run through the hidden fully connected (dense) layer with 64 nodes. To avoid overfitting, a dropout layer was suggested to drop randomly 70% of the learned weights. The final layer is a signle-node layer (output layer).

In most CNN models, the rectified linear unit (ReLU) function is the activation function that is used in all convolutional and fully connected (dense) layers, whereas the sigmoid function is the final activation function used in output nodes. The trials for algorithm run were

	Layers	Output Size	Description
	Batch Size: 16, Epoch	ns: 200, pixel size 256x256x3, dropo	put: 0.7
Layer1	Convolution	(None,256,254,8)	Filters: 8, Kernel Size: 3x3,Input size: 256x256x3, Activation: ReLU
	Pooling	(None, 127, 127,8)	2x2 Max Pooling
	Dropout	(None, 127, 127,8)	Drop Rate = 0.7
Layer2	Convolution	(None, 125, 125,8)	Filters: 16, Kernel Size: 3x3, Activation: ReLU
	Pooling	(None,62,62,16)	2x2 Max Pooling
	Dropout	(None,62,62,16)	Drop Rate = 0.7
Layer3	Convolution	(None,60,60,32)	Filters: 32, Kernel Size: 3x3, Activation: ReLU
	Pooling	(None,30,30,32)	2x2 Max Pooling
	Dropout	(None,30,30,32)	Drop Rate = 0.7
	Flatten	(None,2312)	
	Dense	(None,28800)	64 nodes, Activation: ReLU
	Dropout	(None,64)	Drop Rate = 0.7
	Dense	(None,1)	1 node, Activation: Sigmoid

Table 1. The input vector concepts of each scenario.

https://doi.org/10.1371/journal.pone.0237213.t001

accomplished for different number of epochs (100, 200, 300, 500), aiming at exploiting the most appropriate number of epochs for CNN training. Moreover, two performance metrics, concerning accuracy and loss, are used. As it regards loss, the binary cross-entropy function is calculated with an ADAM optimizer. For model training, the ImageDataGenerator class from Keras was used, offering augmentations on images like rotations, shifting, zoom, flips and more.

4. Results

This section reports on the results produced in this research study, which is devoted to the use of whole-body scans from SPECT, to perform the diagnosis of bone metastasis for patients suffering from prostate cancer. Each process was repeated 10 times to calculate the classification accuracy.

Hardware and software environments

This experiment was performed in a collaboratory environment, called Google Colab [74], which is a free Jupyter notebook environment in the cloud. The main reason for selecting this cloud environment of Google Colab is that supports free GPU acceleration. The frameworks Keras 2.0.2 and TensorFlow 2.0.0. were used, as well as python language 3.7, CNN with structure (Convolution layer, 3; Maxpooling layer, 3) and Adam Optimizer. Other CNN methods, including VGG16 [33], ResNet50 [38], Inception V3 [67], Xception [75], and MobileNet [76] were also implemented, so as a comparative analysis could be performed between them and the method proposed in this paper.

OpenCV was used for loading and manipulating images, Glob for reading filenames from a folder, Matplotlib for plot visualizations and finally, Numpy for all mathematical and array operations. Python was used for coding, with the CNN being programmed with Keras (with Tensorflow [77]), whereas, data normalization, data splitting, confusion matrices and classification reports were carried out with Sci-Kit Learn. The computations ranged between 2' to 4' per training (epoch) for RGB images (256x256x3), depending on the different input.

It is noteworthy that the original images, as acquired from the scanning device, are in RGB format, containing 3-channel color information. Even though the images are in RGB structure,

they appear as grayscale due to the absence of color components in the image. Regarding the type of CNN, we applied a 2D CNN for each color channel of each image, and then aggregated the inputs into the final part of the network, which is comprised by a dense layer.

Following the steps described in section 2.3, the images are initially loaded in RGB mode, by default. Normalization, shuffling and data augmentation are followed just before the training phase. Next, the dataset is split into training, validation and testing and afterwards, images are passed through the proposed CNN network that has been pre-trained on the ImageNet data set [78]. This data set is able to provide another, more efficient method for weights initialization, which seems of high utility, in any image-related task.

The classification task was two-fold, one for bone metastasis presence and another one for metastasis absence in patients suffering from prostate cancer, considering in total, 586 samples of men patients. For the purposes of this study, we employed different CNN-based architectures and hyper-parameter selection, as defined in Section 2.

The whole classification process consisting of these two phases, is illustrated in the following Fig 4, depicting an overall view of the training and testing phase for the examined dataset of 586 patients by classifying them in two categories.

A meticulous CNN exploration process was accomplished, in which we conducted experiments with various convolutional layers, drop rates, epochs, number of dense nodes, pixel sizes and batch sizes. Different values for image pixel sizes were examined, such as 100×100x3, 200×200x3, 256×256x3, 300×300x3 and 350x350x3, as well as various values for batch sizes, such as 8, 16, 32 and 64 were investigated. In addition, variant drop rate values were studied, for example 0.2, 0.5, 0.7 and 0.9 and a divergent number of dense nodes, like 16, 32, 64 and



Fig 4. Overview of training and testing phase of CNN based whole-body image analysis.

128 were explored. The number of epochs that was explored ranged from 100, 200, 300 to 500. The number of convolutional and pooling layers was also investigated.

Notable is the fact that we performed many experiments with several convolutional layers, epochs and pixel sizes, to find the optimum values. We observed that for epochs = 200 and for pixel sizes (256,256,3), the CNN models with three convolutional layers had the best performance concerning the classification accuracy and sensitivity. After a thorough CNN exploration analysis, we concluded that the best CNN configuration was the following: a CNN with 3 convolutional layers, starting with 8 filters for the first layer, and for each convolutional layer that comes next, the number of filters is doubled (8 - > 16 - > 32). The same logic is followed for the following max-pooling layers. All filters have dimensions of 3×3 . This set-up is shown in Table 1. After carefully selecting the CNN architecture along with the number of epochs and pixel sizes (epochs = 200 and pixel = $256 \times 256 \times 3$), we conducted further experiments with varying drop rates and batch sizes, to find out the best performance parameters.

Table 2 gathers the performance analysis and results for the CNN models with 3 convolutional layers (8,16,32), epochs = 200, dropout = 0.7, pixel size = 256x256x3, dense nodes = 64 and different batch sizes (in Google Colab [74]) for 10 runs, whereas Table 3 presents the 5 runs of the best CNN architecture, entailing 3 convolutional layers (8,16,32), epochs = 200, dropout = 0.7, pixel size = 256x256x3, batch size = 16 and different dense nodes.

Table 4 gathers the values of the performance metrics for the set of the most accurate and robust CNN models, which consists of 3 conv (8,16,32), epochs = 200, dropout = 0.7, pixel size = 256x256x3, various dense nodes (32,64,128) for best batch size 16 and various batch sizes for the best number of dense nodes which is 64. Overall, it is observed that the CNN model with batch size 16 and 64 dense nodes has the highest classification accuracy and sensitivity, as well as the minimum loss. S1 Table shows analytically the calculated values of the performance metrics for the best CNN network for 10 runs (batch size = 16, dropout = 0.7).

In what follows, Figs 5 and 6 illustrate the performance of the proposed CNN architecture of 3 convolutional layers, 256x256x3 pixels, 200 epochs and 64 dense nodes, for different drop rates and different batch sizes, respectively. It is clearly observed that the CNN model with the optimum performance, concerning the testing accuracy, is the model with dropout = 0.7 and batch size = 16. In S2, S3 and S4 Tables, we provide some results produced after further simulations for different drop rates and batch sizes for the selected pixel size. Also, some indicative results with different pixel sizes (300, 300, 3) and another efficient CNN architecture,

Table 2.	CNN model with 3	3 conv (8,16,32), epochs =	= 200, dropout = 0.7, pixe	l size = 256x256x3,	dense nodes = 64 and	different batch sizes.
----------	------------------	----------------------------	----------------------------	---------------------	----------------------	------------------------

	batch size = 8				batch size = 16			batch size = 32				
	Acc. Val	Loss Val	Acc Test	Loss Test	Acc. Val	Loss Val	Acc Test	Loss Test	Acc. Val	Loss Val	Acc Test	Loss Test
Run 1	96,88	0,09	95,45	0,12	100,00	0,04	97,50	0,09	95,83	0,12	95,31	0,125
Run 2	95,83	0,09	97,72	0,14	96,88	0,09	96,25	0,08	97,91	0,076	96,875	0,107
Run 3	97,91	0,10	96,59	0,07	94,79	0,15	97,50	0,08	97,91	0,103	96,875	0,103
Run 4	97,92	0,06	96,59	0,08	95,83	0,17	98,75	0,11	94,79	0,126	98,43	0,074
Run 5	96,88	0,11	96,59	0,09	94,79	0,16	98,75	0,06	98,95	0,08	95,31	0,135
Run 6	95,83	0,10	98,86	0,08	97,92	0,10	96,25	0,15	95,83	0,218	90,625	0,293
Run 7	93,75	0,14	96,59	0,17	95,83	0,12	97,50	0,07	97,91	0,073	96,875	0,114
Run 8	94,79	0,11	95,45	0,11	95,83	0,13	97,50	0,05	95,83	0,2	98,43	0,059
Run 9	97,92	0,08	97,72	0,08	98,43	0,07	96,25	0,14	96,87	0,088	96,875	0,107
Run 10	97,92	0,77	95,45	0,09	97,20	0,07	97,50	0,06	98,95	0,077	93,75	0,114
Average	96,56	0,16	96,70	0,10	96,75	0,11	97,38	0,09	97,08	0,12	95,94	0,12

	Dense nodes = 32				Dense nodes = 64			Dense nodes = 128				
	Acc. Val	Loss Val	Acc Test	Loss Test	Acc. Val	Loss Val	Acc Test	Loss Test	Acc. Val	Loss Val	Acc Test	Loss Test
Run 1	97,92	0,08	95,00	0,16	100,00	0,04	97,50	0,09	97,92	0,1	98,75	0,05
Run 2	96,87	0,12	97,50	0,09	96,88	0,09	96,25	0,08	98,96	0,05	97,5	0,09
Run 3	98,96	0,06	97,50	0,11	94,79	0,15	97,50	0,08	96,87	0,09	96,25	0,09
Run 4	96,87	0,07	98,75	0,09	95,83	0,17	98,75	0,11	97,92	0,05	98,75	0,06
Run 5	97,92	0,06	96,25	0,13	94,79	0,16	98,75	0,06	95,83	0,12	93,75	0,14
Average	97,71	0,08	97,00	0,12	96,46	0,12	97,75	0,08	97,5	0,08	97	0,09

Table 3. CNN model with 3 conv (8,16,32), epochs = 200, dropout = 0.7, pixel = 256x256x3, different dense nodes and batch size = 16.

https://doi.org/10.1371/journal.pone.0237213.t003

consisting of four convolutional layers (8,16,32,64), epochs = 200, dropout = 0.7, pixel = 256x256x3, various dense nodes and the best batch size 16 are depicted in <u>\$5</u> and <u>\$6</u> Tables, respectively.

For the best dropout rate (0.7) and the best batch size (16), the accuracies for various dense nodes (32, 64 and 128) are depicted in Table 3. It is observed that these CNN models, with dense nodes 32, 64 and 128, provide similar classification accuracies, with minimum testing loss. The most accurate CNN model with respect to classification accuracy, sensitivity and loss, is the CNN with dropout 0.7, batch size 16 and 64 dense nodes. The best confusion matrix for the proposed CNN model, considering malignant (metastasis) and benign (no metastasis) patients, is shown in Table 5.

Through this extensive CNN exploration, it is summarized that the best CNN model has the following characteristics: batch-size = 16, dropout = 0.7, nodes (3 conv layers) = 8, 16, 32, Dense Nodes = 64, Epochs = 200 and Pixel = (256, 256, 3). Fig 7 represents the precision curves of testing accuracy and loss for the CNN model that performed the best, whereas Fig 8 illustrates the precision curves for the other two CNN architectures, with best dropout = 0.7 and best batch size 16, considering 32 and 128 dense nodes, respectively.

To further investigate the performance of the proposed CNN architecture, an extensive comparative analysis between the state of the art CNNs, such as VGG16 [43, 44], ResNet50

		Batch size = 16			Dense nodes = 64			
			Dense nodes		Batcl	Batch size		
		32	64	128	8	32		
Accuracy		97	97.75	97	96.70	95.94		
Loss		0,12	0.08	0,09	0.10	0.12		
Precision	Malignant	0.966	0.987	0.872	0.98	0.96		
	Healthy	0.976	0.947	0.85	0.947	0.972		
Recall	Malignant	0.984	0.973	0.867	0.966	0.974		
	Healthy	0.946	0.975	0.854	0.965	0.94		
F1-Score	Malignant	0.976	0.98	0.87	0.972	0.966		
	Healthy	0.958	0.96	0.848	0.955	0.954		
Sensitivity	Malignant	0.968	0.984	0.872	0.98	0.959		
	Healthy	0.974	0.945	0.847	0.946	0.966		
Specificity	Malignant	0.974	0.95	0.849	0.946	0.97		
	Healthy	0.968	0.987	0.872	0.981	0.932		
Execution Time (s)		905	958	979	1008	927		

Table 4. Evaluation metrics for the set of CNN models consists of 3 conv (8,16,32), epochs = 200, dropout = 0.7, pixel size = 256x256x3, various dense nodes and batch sizes.



Fig 5. Bar chart with calculated accuracies for various batch sizes (8, 16, 32, 64) and dropout = 0.7.

https://doi.org/10.1371/journal.pone.0237213.g005

[38], Inception V3 [67], Mobile Net [76] and Xception [75] and our best model, was performed. The following well-known CNN architectures were used: (i) ResNet50, which is a 50 weight layer deep version of ResNet (Residual neural Network), with 152 layers based on "network-in-network" micro-architectures [38]. ResNet has less parameters than the VGG network, demonstrating that extremely deep networks can be trained using standard SGD (and a reasonable initialization function) through the use of residual modules. (ii) VGG16 [43, 44], which is an extended version of VGG (Visual Geometry Group) as it contains 16 weight layers within the architecture. VGGs are usually constructed by using 3×3 convolutional layers, which are stacked on top of each other.

Inception V3 [67], which is Inception's third installment, includes new factorization ideas. It is a 48-layers deep network which incorporates RMSProp optimizer and computes 1×1 , 3×3 , and 5×5 convolutions within the same module of the network. Its original architecture is GoogleNet. Szegedy et al. proposed an updated version of the architecture of Inception V3, which was included in Keras' inception module [67]. This updated version is capable to further boost





	Malignant	Benign
Malignant	61	0
Benign	1	26
-	1	

Table 5. Best confusion matrix for the proposed CNN.

https://doi.org/10.1371/journal.pone.0237213.t005

the classification accuracy of ImageNet. Xception, which is an extension of the Inception architecture, replaces the standard Inception modules with depth-wise separable convolutions [75]. MobileNet convolutes each channel separately instead of combining and flattening them all, with the use of depth-wise separable convolutions [76]. Its architecture combines convolutional layers, depth-wise and point-wise layers to a total number of 30. These popular CNNs were used for transfer learning with the weights from the ImageNet [78] dataset.

In this research work, after an extensive exploration with the provided architectures of popular CNNs, the following parameters were defined, regarding the well-known CNNs. Table 6 depicts the optimum parameters for the five well-known CNNs (VGG16, ResNet50, MobileNet, Inception V3 and Xception), compared with the proposed best performed CNN. Average running time was calculated for each CNN architecture for all models and is also presented in Table 6. It is obvious that the proposed CNN model is less time-consuming, providing at the same time higher classification accuracy, when compared with all the other popular CNN models.

In what follows, Tables 7, 8 and 9 gather the results of the state-of-the-art CNN models, which are straightforward compared with our best performed CNN configuration, suggested in this research work.



Fig 7. Precision curves for best CNN architecture with dense = 64, dropout = 0.7, in RGB mode: (a) Accuracy and (b) loss.

https://doi.org/10.1371/journal.pone.0237213.g007







CNNs	Epochs, pixel size, dropout, batch size, dense nodes,	Average Running Time for 200 epochs
Best CNN	Batch-size = 16, dropout = 0.7, nodes [3 conv layers) = 8,16,32, Dense Nodes: 64, Epochs:200, Pixel = [256x256x3)	418 sec
VGG16	Pixel size (256x256x3), batch size = 64, dropout = 0.2, dense nodes 2x512, epochs = 200	1005 sec
ResNet50	Pixel size (300x300x3), batch size = 8, dropout = 0.2, Global Average Pooling, dense nodes 512x512, epochs = 200	2434 sec
MobileNet	Pixel size (300x300x3), batch size = 16, dropout = 0.2, global average pooling, epochs = 200	1380 sec
Inception V3	Pixel size (250x250x3), batch size = 16, dropout = 0.7, dense nodes = 1500x1500, epochs = 200	1317 sec
Xception	Pixel size (300x300x3), batch size = 8, dropout = 0.7, Global Average Pooling, dense nodes = 512x512, epochs = 200	3532 sec

Table 6. Selected values of parameters for benchmark CNNs.

https://doi.org/10.1371/journal.pone.0237213.t006

It is worth mentioning that we have conducted an exploratory analysis for all the benchmark CNNs, with different dropouts, while reducing the number of weights (to find a network complexity appropriate for the problem) to avoid overfitting [25]. Thus, we selected not only the best performing model of each CNN architecture concerning the accuracy, but at the same time, the best CNN which avoids overfitting in 10 runs. Indicative results of the five state-ofthe-art CNN models for the selected optimum parameters, pixels, dropouts and batch sizes are gathered in S7 and S8 Tables. Furthermore, in S9 Table, the best confusion matrices for the five benchmark CNN models are gathered. Fig 9 illustrates the average prediction accuracy with the respective loss curve for the best performed state of the art CNN (MobileNet).

5. Discussion of results

The problem of diagnosis of bone metastasis in prostate cancer patients has been tackled with the use of CNN algorithms, which are considered applicable and powerful methods for detecting complex visual patterns in the field of medical image analysis. In the current work, a total of 586 images was acquired, that includes similar numbers of both healthy and malignant cases from metastatic prostate cancer patients. Several CNN architectures were tested, leading to the one that performed optimum under all hyperparameter selection cases and regularization methods, emerging classification accuracies ranging from 93.75% to 98.685%.

After a thorough exploration analysis, as reported in section 3, a CNN architecture with the following specifications: 3 convolutional layers, batch size = 16, dropout = 0.7 and dense nodes = 64, has provided the highest classification accuracy (97.38%) and the lowest loss (0.08) among all the architectures investigated in this work. Concerning the evaluation metrics for the proposed model configuration, as depicted in Table 4, it is observed that precision (0.969), recall (0.974), sensitivity (0.965), specificity (0.968) and F1 score (0.97) present the highest values when compared with those of the rest of the examined model configurations. It is worth

Table 7. A	Accuracy and	Loss (validation an	d testing) of th	e proposed CN	N model and bench	ımark CNNs.
------------	--------------	---------------------	------------------	---------------	-------------------	-------------

Average	Proposed CNN	ResNet50	VGG16	MobileNet	Inception V3	Xception
Accuracy (Validation)	96.04	95	93.44	97.29	95.52	97.5
Loss (Validation)	0.13	0.12	0.17	0.07	0.12	0.08
Accuracy (Testing)	97.38	94.98	93.75	98.13	96.7	96.36
Loss (Testing)	0.087	0.16	0.17	0.06	0.14	0.13

Network	Proposed CNN	ResNet50	VGG16	MobileNet	Inception V3	Xception
Accuracy	97,38	94.98	93.75	98.13	96.7	96.36
Loss	0.087	0.16	0.17	0.06	0.14	0.13
Precision	0.987	0.938	0.922	0.986	0.951	0.942
Recall	0.973	0.962	0.935	0.981	0.97	0.99
F1-score	0.98	0.948	0.922	0.983	0.958	0.968
Sensitivity	0.984	0.939	0.921	0.986	0.952	0.941
Specificity	0.95	0.937	0.869	0.965	0.955	0.99

Table 8. Malignant disease class performance comparison of different methods.

https://doi.org/10.1371/journal.pone.0237213.t008

mentioning that when it comes to the proposed architecture with a different number of dense nodes (32 or 128), it is proven that the produced classification model is still considerably accurate even though it provides a lower testing accuracy (97%) than that of the proposed CNN model.

To further discuss the results produced, we compare (even indirectly) the classification performance of our model with that of other ML algorithms and CNNs, which were reported in the literature and already applied in the particular problem of bone metastasis classification in nuclear medicine. Due to the fact that none of these research studies provide publicly available data, it is not feasible to accomplish a straightforward comparison with them. Thus, we gathered all the related research works in this domain and present them in Table 10 (as mentioned in the Introduction section), accompanied with the calculated classification accuracies and evaluation metrics, like sensitivity and specificity.

A comparative analysis has been further conducted by authors to demonstrate the novelty of the proposed method. This analysis concerns recent ML and CNN techniques presented in related works in bone scintigraphy imaging, along with their specifications, as listed in Table 10. As regards the previous works of [49] and [50] in bone scintigraphy classification, they employ CNNs for metastatic / non-metastatic hotspots classification. They are highly related to this research study and provide classification accuracies up to 89%. Taking into consideration only the CNN methods with 2 classes (the same number of classes as the proposed model), then the highest possible accuracy value (95.9%) is achieved in [62]. Overall, comparing the classification accuracy value of the proposed CNN method (97.38%) with those of the reported methods listed in Table 10, authors come to the conclusion that the proposed CNN approach outperforms all the previous ML and deep learning techniques in bone scintigraphy imaging.

Besides accuracy, several other evaluation metrics like precision, recall, sensitivity, specificity and F1 score, as listed in Tables 7, 8 and 9, were used by authors to measure the classification performance of the proposed model as well as other state-of-the-art CNNs, commonly used for image classification problems. In regard to the testing accuracy values for the specific

Network	Proposed CNN	ResNet50	VGG16	MobileNet	Inception V3	Xception
Accuracy	97.38	94.98	93.75	98.13	96.7	96.36
Precision	0.947	0.938	0.87	0.966	0.954	1
Recall	0.975	0.9	0.832	0.973	0.908	0.872
F1-score	0.96	0.912	0.835	0.968	0.927	0.926
Sensitivity	0.945	0.937	0.921	0.973	0.953	1
Specificity	0.987	0.939	0.921	0.983	0.953	0.941

Table 9. Healthy class performance comparison of different methods.



Fig 9. Precision curves for MobileNet (a) accuracy and (b) loss.

https://doi.org/10.1371/journal.pone.0237213.g009

problem, the suggested algorithm seems to outperform all the reported benchmark CNNs (their testing accuracy is up to 96.7%), except from MobileNet, whose accuracy results (98.13%) are slightly higher than that of the proposed approach. The produced values for the rest of the reported metrics of all the examined architectures present a similar picture, thus, concluding that our model exhibits better or similar performance to the other state-of-the-art CNN architectures. This study validates the premise that CNNs are algorithms that can offer high accuracy in medical image classification-based problems. This has a direct application on medical imaging where the automatic identification of diseases is crucial for the patients. The main outcomes of this study can be summarized as follows:

- i. The proposed CNN method exhibits outstanding performance when RGB analysis is performed for the examined images of this case study. This can be emanated from the fact that the results produced after the application of CNN architecture are based on distinct features, that appear specifically on the bone metastasis presence scans, compared to the healthy scans (no malignant spots).
- ii. The proposed CNN architecture is superior to four out of five benchmark and well-known CNN architectures (ResNet50, VGG16, Inception V3 and Xception), which have been efficiently used in medical image processing problems. As it is observed from Tables 8, 9 and 10, the results deriving from the application of the CNN method, are better in terms of classification accuracy, prediction, sensitivity and F1 score, than those coming from popular

and well-known CNN approaches, found in the relevant literature. In the case of Mobile-Net, it shows a relatively similar classification accuracy to the proposed model.

- iii. The proposed CNN models appear to have significant potential since they exhibit better performance with less running time and simpler architecture than other well-known CNN architectures, considering the case of bone metastasis classification of whole-body scans.
- iv. The proposed bone scan CNN performs efficiently, despite the fact that it was trained on a small number of images.
- v. Overall, for the purpose of medical image analysis and classification, the CNN methodology is proven to be powerful enough in the nuclear medicine domain and particularly, for bone scintigraphy, outweighing the popular CNN architectures for image analysis, like VGG16, ResNet50, MobileNet, InceptionV3 and Xception.

6. Conclusions and future work

It is evident that medical imaging paired with machine learning techniques, specifically with CNNs, is considerably valuable for clinical diagnosis of bone metastasis. Considering and exploiting all new enhancements in the field of deep learning, a simpler, faster and more accurate set of CNN networks for classification is proposed. In the case where only bone scan images are used at the input level, the classifier can identify the presence of bone metastasis in prostate cancer patients. The proposed CNN-based method in this work, outperforms all other popular and previously proposed CNN methods, achieving excellent performance in terms of classification accuracy, precision, recall, sensitivity and specificity indicators. Accordingly, the

	Reference	Year	ML Methods	Classification problem	Results
Bone scintigraphy	Sadik et al. [<u>51</u>]	2008	ANNs	Metastasis present or absent	Sensitivity 90% and accuracy 74%
(bone scans)	Wang et al. [58]	2016	DT and SVM	(2 classes) predict metastasis or not	87.58% ± 2.25%
	Dang J. (Master Thesis) [<u>49</u>]	2016	CNNs	(2 classes) metastatic / non-metastatic hotspots (*)	calculated accuracy of the testing set 0.89
	Belcher L. (Master Thesis) [50]	2017	CNNs	(2 classes) metastatic / non-metastatic hotspots	AUC score 0.937
	Horikoshi et al. [<u>48</u>]	2012	ANNs	metastatic disease or not	Sensitivity 90% and accuracy 83%
	Aslantas et al. [<u>56</u>]	2016	CADBOSS (ANNs)	(2 classes) Metastasis present or absent	The accuracy, sensitivity, and specificity were 92.30%, 94%, and 86.67%
FDG PET or FDG PET-CT	Bradshaw T/ et al. [59]	2018	CNNs	(2 classes) classification of benign and malignant bone lesions	0.88, 0.90, 0.85, and 0.90, concerning the accuracy, sensitivity, specificity & positive predictive value
	Furuya S. et al. [<u>60]</u>	2019	Deep Learning	(3 classes) 1) benign, 2) malignant, and 3) equivocal	Accuracies for benign, malignant and equivocal were 99.4%, 99.4% and 87.5%, respectively
	Furuya S. et al. [<u>61</u>]	2019	ResNet24	(3 classes) 1) benign in the head-and-neck region, 2) malignant in the head-and-neck region, and 3) equivocal in the head-and-neck region	Accuracies for 1), 2) and 3) were 97.3%, 97.8% and 96.2% respectively.
	Kawauchi K et al. [62]	2019	CNN	(2 classes) classifying these patients by sex (male- female)	The accuracy values were 95.9% for male and 90.2% for female.
	Kawauchi K et al. [63]	2018	CNNs	(2 classes) predict whether physician's further diagnosis is required or not	Accuracy 93.2±3.9%

Table 10. Related works in bone scintigraphy imaging for prostate cancer metastasis using ML and CNN techniques.

 * Only the hotspots found in the spine had been used to train the CNN.

effectiveness of the suggested improvements to the network architecture and hyperparameter configuration are presented in this research study. The experimental results demonstrate that the proposed method achieves outstanding performance comparing with other methods that make use of convolutional neural networks. In particular, this approach allows an easier, faster and more precise interpretation of scintigraphy images, which can have a positive impact on diagnosis accuracy as well as on decision making, regarding the treatment that will be further administered.

Even though this CNN approach uses a relatively small dataset of patients, this work suggests that bone scintigraphy, combined with the CNN trained models, can have a considerable effect in the detection of bone metastasis. In our next step, more images will be gathered regarding patients suffering from prostate cancer, as well as patients suffering from other types of metastatic cancer, like breast, kidney, lung or thyroid cancer in order to thoroughly investigate the proposed architecture. The results seem quite promising and really encouraging when computer aided diagnosis is concerned, making the proposed network more useful for clinical routine work.

Summarizing, future plans include further investigation of both deep learning methods and the proposed architecture in two directions; to collect more bone scans from patients suffering from various types of metastatic cancer (except prostate cancer) and to examine interpretability of the model and workflow integration.

Supporting information

S1 Appendix. Convolutional neural networks basics. (DOCX)

S2 Appendix. Classification performance metrics. (DOCX)

S1 Table. Performance metrics for the best CNN network for 10 runs (batch size = 16, dropout = 0.7). (DOCX)

S2 Table. CNN model 1 (epochs = 200, dropout = 0.2, pixel = 256x256x3), and dense nodes = 64. (DOCX)

S3 Table. Model 1 (epochs = 200, dropout = 0.5, pixel = 256x256x3) dense nodes = 64. (DOCX)

S4 Table. CNN Model 1 (epochs = 200, dropout = 0.9, pixel = 256x256x3) dense nodes = 64. (DOCX)

S5 Table. CNN model (epochs = 200, dropout = 0.7, pixel = 300x300x3) runs for different batch sizes.

(DOCX)

S6 Table. CNN model with 4 conv (8,16,32,64), epochs = 200, dropout = 0.7, pixel = 256 x 256 x 3, different dense nodes and batch size = 16. (DOCX)

S7 Table. 10 Runs for VGG16 (epochs = 200, dropout = 0.7, pixel = 256 × 256x3), resNet50 (epochs = 200, dropout = 0.2, pixel = 300 × 300 x 3) and MobileNet (epochs = 200, dropout = 0.2, pixel = 300 × 300 x 3). (DOCX)

S8 Table. 10 Runs for inceptionV3 (epochs = 200, dropout = 0.7, pixel = $250 \times 250 \times 3$, dense = 2x1500) and xception (epochs = 200, dropout = 0.7, pixel = $300 \times 300 \times 3$, dense = 2x512). (DOCX)

S9 Table. Confusion matrices for state of the art CNN models. (DOCX)

Acknowledgments

Authors would like to express their sincere thanks and appreciation to master student Ms. Anna Feleki for her involvement in this work.

Author Contributions

Conceptualization: Nikolaos Papandrianos, Elpiniki Papageorgiou.

Data curation: Nikolaos Papandrianos.

Formal analysis: Nikolaos Papandrianos, Elpiniki Papageorgiou.

Investigation: Nikolaos Papandrianos.

Methodology: Nikolaos Papandrianos, Elpiniki Papageorgiou, Athanasios Anagnostis.

Project administration: Elpiniki Papageorgiou.

Resources: Nikolaos Papandrianos, Elpiniki Papageorgiou, Konstantinos Papageorgiou.

Software: Elpiniki Papageorgiou, Athanasios Anagnostis.

Supervision: Nikolaos Papandrianos, Elpiniki Papageorgiou.

Validation: Nikolaos Papandrianos, Elpiniki Papageorgiou, Athanasios Anagnostis.

Visualization: Konstantinos Papageorgiou.

Writing - original draft: Nikolaos Papandrianos.

Writing – review & editing: Nikolaos Papandrianos, Elpiniki Papageorgiou, Athanasios Anagnostis, Konstantinos Papageorgiou.

References

- 1. Lukaszewski B, Nazar J, Goch M, \Lukaszewska M, Stępiński A, Jurczyk MU. Diagnostic methods for detection of bone metastases. 2017.
- Mundy GR. Metastasis to bone: causes, consequences and therapeutic opportunities. Nat Rev Cancer. 2002; 2:584–593. https://doi.org/10.1038/nrc867 PMID: 12154351
- Mingyu Z, Xin L, Yuan Q, Silong H, Yingjian Z, Wentao L, et al. Bone metastasis pattern of cancer patients with bone metastasis but no visceral metastasis. J Bone Oncol [Internet]. 2019 Apr; 15. Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC6357895/
- Sartor O, DiBiase SJ. MD, Bone metastases in advanced prostate cancer: Clinical manifestations and diagnosis. 2018; Available from: https://www.uptodate.com/contents/bone-metastases-in-advancedprostate-cancer-clinical-manifestations-and-diagnosis
- Sartor AO, DiBiase SJ. Bone metastases in advanced prostate cancer: Management. 2018; Available from: https://www.uptodate.com/contents/bone-metastases-in-advanced-prostate-cancermanagement
- Rieden K. Conventional imaging and computerized tomography in diagnosis of skeletal metastases. Radiologe. 1995; 35:15–20. PMID: 7892438

- Hamaoka T, Madewell JE, Podoloff DA, Hortobagyi GN, Ueno NT. Bone imaging in metastatic breast cancer. Journal of Clinical Oncology. 2004;
- Even-Sapir E, Metser U, Mishani E, Lievshitz G, Lerman H, Leibovitch I. The detection of bone metastases in patients with high-risk prostate cancer: 99mTc-MDP Planar bone scintigraphy, single- and multifield-of-view SPECT, 18F-fluoride PET, and 18F-fluoride PET/CT. Journal of Nuclear Medicine. 2006 Feb; 47(2):287–97. PMID: 16455635
- Gold L, Lee C, Devine B. Imaging techniques for treatment evaluation for metastatic breast cancer. Rockville (MD): Agency for Healthcare Research and Quality (US). 2014 Oct; <u>https://doi.org/10.1007/s12282-014-0560-0</u>
- 10. Fogelman I, Cook G, Israel O, Van Der Wall H. Positron emission tomography and bone metastases. Seminars in Nuclear Medicine Elsevier. 2005; 35:135–42.
- Pianou NK, Stavrou PZ, Vlontzou E, Rondogianni P, Exarhos DN, Datseris E. More advantages in detecting bone and soft tissue metastases from prostate cancer using 18F-PSMA PET/CT. Hell J Nucl Med. 2019 Apr; 22(1):6–9. https://doi.org/10.1967/s002449910952 PMID: 30843003
- Newberg A. Chapter 54—Bone Scans. In: Radiology Secrets Plus (Third Edition) [Internet]. 2011. p. 376–81. Available from: https://doi.org/10.1016/B978-0-323-06794-2.00054-7
- Wyngaert T, Strobel K, Kampen WU, Kuwert T, Bruggen W, Mohan HK, et al. On behalf of the EANM Bone & Joint Committee and the Oncology Committee. The EANM practice guidelines for bone scintigraphy. Eur J Nucl Med Mol Imaging. 2016; 43:1723–38. <u>https://doi.org/10.1007/s00259-016-3415-4</u> PMID: 27262701
- Drakopoulos M, Vejdani K, Osman M. Comparison of diagnostic certainty and accuracy of 18F-NaF PET/CT and planar 99mTc-MDP bone scan in patients with prostate cancer. J Nucl Med. 2014 May; 55 (supplement 1 1669).
- Litjens G, Kooi T, Bejnordi BE, Setio AAA, Ciompi F, Ghafoorian M, et al. A survey on deep learning in medical image analysis. Medical Image Analysis. 2017; 42:60–88. https://doi.org/10.1016/j.media. 2017.07.005 PMID: 28778026
- Biswas M, Kuppili V, Saba L, Edla DR, Suri HS, Cuadrado-Godia E, et al. State-of-the-art review on deep learning in medical imaging. Frontiers in bioscience (Landmark edition) [Q1]. 2019 Jan; 24:392– 426.
- 17. Abdelhafiz D, Yang C, Ammar R, Nabavi S. Deep convolutional neural networks for mammography: Advances, challenges and applications. BMC Bioinformatics. 2019 Jun; 20(11):281.
- Lundervold AS, Lundervold A. An overview of deep learning in medical imaging focusing on MRI. Zeitschrift für Medizinische Physik. 2019 May; 29(2):102–27. <u>https://doi.org/10.1016/j.zemedi.2018.11.002</u> PMID: 30553609
- Komeda Y, Handa H, Watanabe T, Nomura T, Kitahashi M, Sakurai T, et al. Computer-Aided Diagnosis Based on Convolutional Neural Network System for Colorectal Polyp Classification: Preliminary Experience. Oncology (Switzerland). 2017; 93(1):30–4.
- Shen D, Wu G, Suk H-I. Deep Learning in Medical Image Analysis. Annual Review of Biomedical Engineering. 2017; 19:221–48. https://doi.org/10.1146/annurev-bioeng-071516-044442 PMID: 28301734
- Xue Y, Chen S, Qin J, Liu Y, Huang B, Chen H. Application of Deep Learning in Automated Analysis of Molecular Images in Cancer: A Survey. Contrast Media Mol Imaging. 2017 Oct 15;9512370. <u>https://doi.org/10.1155/2017/9512370</u> PMID: 29114182
- 22. O'Shea KT, Nash R. An Introduction to Convolutional Neural Networks. arXiv preprint [Internet]. 2015; Available from: https://arxiv.org/abs/1511.08458
- Kahn CE Jr. From Images to Actions: Opportunities for Artificial Intelligence in Radiology. Radiology. 2017; 285:719–20. https://doi.org/10.1148/radiol.2017171734 PMID: 29155645
- Dreyer KJ, Geis JR. When Machines Think: Radiology's Next Frontier. Radiology. 2017; 285:713–8. https://doi.org/10.1148/radiol.2017171183 PMID: 29155639
- Yamashita R, Nishio M, Do RKG, Togashi K. Convolutional neural networks: an overview and application in radiology. Insights into Imaging. 2018 Aug; 9(4):611–29. <u>https://doi.org/10.1007/s13244-018-0639-9 PMID: 29934920</u>
- Brinkerab T. Deep learning outperformed 136 of 157 dermatologists in a head-to-head dermoscopic melanoma image classification task. European Journal of Cancer. 2019 May; 113:47–54. <u>https://doi.org/10.1016/j.ejca.2019.04.001</u> PMID: 30981091
- 27. Yang L, Xie X, Li P, Zhang D, Zhang L. Part-based convolutional neural network for visual recognition. In: Image Processing. IEEE;
- Alaskar H, Hussain A, Al-Aseem N, Liatsis P, Al-Jumeily D. Application of Convolutional Neural Networks for Automated Ulcer Detection in Wireless Capsule Endoscopy Images. Sensors (Basel). 2019 Mar; 19(6):1265.

- Jiang Y, Chen L, Zhang H, Xiao X. Breast cancer histopathological image classification using convolutional neural networks with small SE-ResNet module. PLoS One. 2019 Mar 29; 14(3):e0214587. https:// doi.org/10.1371/journal.pone.0214587 PMID: 30925170
- Lakhani P, Sundaram B. Deep Learning at Chest Radiography: Automated Classification of Pulmonary Tuberculosis by Using Convolutional Neural Networks. Radiology. 2017; 284:574–582. https://doi.org/ 10.1148/radiol.2017162326 PMID: 28436741
- Li Z, Wang Y, Yu J, Guo Y, Cao W. Deep Learning based Radiomics (DLR) and its usage in noninvasive IDH1 prediction for low-grade glioma. Sci Rep. 2017; 7:5467. https://doi.org/10.1038/s41598-017-05848-2 PMID: 28710497
- Khened M, Kollerathu V, Krishnamurthi G. Fully convolutional multi-scale residual DenseNets for cardiac segmentation and automated cardiac diagnosis using ensemble of classifiers. Medical Image Analysis. 2018; 51:21–45. https://doi.org/10.1016/j.media.2018.10.004 PMID: 30390512
- Xu L. Automated Whole-Body Bone Lesion Detection for Multiple Myeloma on (68)Ga-Pentixafor PET/ CT Imaging Using Deep Learning Methods. Contrast Media Mol Imaging. 2018;
- Blanc-Durand P, Van Der Gucht A, Schaefer N, Itti E, Prior J. Automatic lesion detection and segmentation of 18F-FET PET in gliomas: A full 3D U-Net convolutional neural network study. PLoS ONE. 2018; 13.
- Yasaka K, Akai H, Abe O, Kiryu S. Deep Learning with Convolutional Neural Network for Differentiation of Liver Masses at Dynamic Contrast-enhanced CT: A Preliminary Study. Radiology. 2018; 286:887– 896. https://doi.org/10.1148/radiol.2017170706 PMID: 29059036
- Ronneberger O, Fischer P, Brox T. U-net: Convolutional networks for biomedical image segmentation. In Springer; 2015. p. 234–241.
- Szegedy C, Liu W, Jia Y, Sermanet P, Reed S, Anguelov D, et al. Going deeper with convolutions. In: Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition. 2015. p. 1–9.
- He K, Zhang X, Ren S, Sun J. Deep residual learning for image recognition. In: Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition. 2016. p. 770–8.
- Chen LC, Papandreou G, Kokkinos I, Murphy K, Yuille AL. DeepLab: Semantic Image Segmentation with Deep Convolutional Nets, Atrous Convolution, and Fully Connected CRFs. IEEE Transactions on Pattern Analysis and Machine Intelligence. 2018;
- Krizhevsky A, Sutskever I, Hinton GE. ImageNet classification with deep convolutional neural networks. In: Advances in neural information processing systems. Curran Associates, Inc.; 2012. p. 1097–105.
- Lecun Y, Bottou L, Bengio Y, Haffner P. Gradient-based learning applied to document recognition. In: IEEE Proceedings [Internet]. 1989. Available from: <u>http://yann.lecun.com/exdb/publis/pdf/lecun-01a.</u> pdf
- Zeiler MD, Fergus R. Visualizing and Understanding Convolutional Networks BT—Computer Vision– ECCV 2014. In: European Conference on Computer Vision (ECCV). Springer; 2014. p. 818–33.
- **43.** Simonyan K, Zisserman A. Very deep convolutional networks for large-scale image recognition. In: 3rd International Conference on Learning Representations, ICLR 2015—Conference Track Proceedings. arXiv; 2014. p. 1409–556.
- 44. Simonyan K, Zisserman A. VGG-16. arXiv preprint. 2014;
- 45. Huang G, Liu Z, Van Der Maaten L, Weinberger KQ. Densely connected convolutional networks. In: Proceedings - 30th IEEE Conference on Computer Vision and Pattern Recognition, CVPR 2017. 2017.
- Erdi YE, Humm JL, Imbriaco M, Yeung H, Larson SM. Quantitative bone metastases analysis based on image segmentation. Journal of Nuclear Medicine. 1997; 38(9):1401–6. PMID: 9293797
- 47. Imbriaco M, Larson SM, Yeung HW. A new parameter for measuring metastatic bone involvement by prostate cancer: the Bone Scan Index. Clin Cancer Res. 1998; 4(7):1765–72. PMID: 9676853
- Horikoshi H, Kikuchi A, Onoguchi M, Sjöstrand K, Edenbrandt L. Computer-aided diagnosis system for bone scintigrams from Japanese patients: importance of training database. Annals of Nuclear Medicine. 2012 Oct; 26(8):622–626. https://doi.org/10.1007/s12149-012-0620-5 PMID: 22729550
- 49. Dang J. Classification in Bone Scintigraphy Images Using Convolutional Neural Networks. 2016. (Master's Theses in Mathematical Sciences).
- Belcher L. Convolutional Neural Networks for Classification of Prostate Cancer Metastases Using Bone Scan Images. 2017.
- Sadik M, Hamadeh I, Nordblom P, Suurkula M, Höglund P, Ohlsson M, et al. Computer-assisted interpretation of planar whole-body bone scans. Journal of Nuclear Medicine. 2008; 49(12):1958–65. https://doi.org/10.2967/jnumed.108.055061 PMID: 18997038

- Sadik M, Suurkula M, Hoglund P, Jarund A, Edenbrandt L. Quality of planar whole-body bone scan interpretations: a nationwide survey. Eur J Nucl Med Mol Imaging. 2008; 35:1464–72. https://doi.org/ 10.1007/s00259-008-0721-5 PMID: 18373092
- Sadik M, Jakobsson D, Olofsson F, Ohlsson M, Suurkula M, Edenbrandt L. A new computer-based decision-support system for the interpretation of bone scans. Nucl Med Commun. 2006; 27:417–23. https://doi.org/10.1097/00006231-200605000-00002 PMID: 16609352
- Sadik M. Computer-Assisted Diagnosis for the Interpretation of Bone Scintigraphy: A New Approach to Improve Diagnostic Accuracy. Gothenburg, Sweden: University of Gothenburg; 2009. (PhD Thesis).
- 55. Tokuda O, Harada Y, Ohishi Y, Matsunaga N, Edenbrandt L. Investigation of computer-aided diagnosis system for bone scans: A retrospective analysis in 406 patients. Ann Nucl Med. 2014; 28:329–39. https://doi.org/10.1007/s12149-014-0819-8 PMID: 24573796
- Aslantaş A, Dandil E, Sağlam S, Çakiroğlu M. CADBOSS: A computer-aided diagnosis system for whole-body bone scintigraphy scans. Journal of Cancer Research and Therapeutics. 2016;
- Elfarra FG, Calin MA, Parasca SV. Computer-aided detection of bone metastasis in bone scintigraphy images using parallelepiped classification method. Ann Nucl Med. 2019 Nov; 33(11):866–74. <u>https://</u> doi.org/10.1007/s12149-019-01399-w PMID: 31493203
- Wang Z, Wen X, Lu Y, Yao Y, Zhao H. Exploiting machine learning for predicting skeletal-related events in cancer patients with bone metastases. Oncotarget. 2016 Mar 15; 7(11):12612–22. https://doi.org/10. 18632/oncotarget.7278 PMID: 26871471
- Bradshaw T, Perk T, Chen S, Hyung-Jun I, Cho S, Perlman S, et al. Deep learning for classification of benign and malignant bone lesions in [F-18]NaF PET/CT images. J Nucl Med. 2018; 59(S1):327.
- Furuya S, Kawauchi K, Hirata K, Manabe W, Watanabe S, Kobayashi K, et al. A convolutional neural network-based system to detect malignant findings in FDG PET-CT examinations. J Nucl Med. 2019 May 1; 60(1):1210.
- Furuya S, Kawauchi K, Hirata K, Manabe W, Watanabe S, Kobayashi K, et al. Can CNN detect the location of malignant uptake on FDG PET-CT? J Nucl Med. 2019 May 1; 60(S1):285.
- Kawauchi K, Hirata K, Katoh C, Ichikawa S, Manabe O, Kobayashi K, et al. A convolutional neural network-based system to prevent patient misidentification in FDG-PET examinations. Scientific Reports. 2019; 9:7192. https://doi.org/10.1038/s41598-019-43656-y PMID: 31076620
- Kawauchi K, Hirata K, Ichikawa S, Manabe O, Kobayashi K, Watanabe S, et al. Strategy to develop convolutional neural network-based classifier for diagnosis of whole-body FDG PET images. J Nucl Med. 2018 May 1; 59(S1):326.
- **64.** Gjertsson K. Segmentation in Skeletal Scintigraphy Images using Convolutional Neural Networks. 2017. (Master's Theses in Mathematical Sciences).
- 65. Leung K, Ashrafinia S, Salehi Sadaghiani M, Dalaie P, Tulbah R, Yin Y, et al. A fully automated deeplearning based method for lesion segmentation in 18F-DCFPyL PSMA PET images of patients with prostate cancer. Journal of Nuclear Medicine. 2019; 60(supplement 1):399.
- 66. Rosebrock A. ImageNet: VGGNet, ResNet, Inception, and Xception with Keras [Internet]. ImageNet: VGGNet, ResNet, Inception, and Xception with Keras. 2017. Available from: https://www. pyimagesearch.com/2017/03/20/imagenet-vggnet-resnet-inception-xception-keras/
- Szegedy C, Vanhoucke V, loffe S, Shlens J, Wojna Z. Rethinking the Inception Architecture for Computer Vision. In: Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition. 2016.
- 68. Weiner MG, Jenicke L, Mller V, Bohuslavizki HK. Artifacts and nonosseous, uptake in bone scintigraphy. Imaging reports of 20 cases. Radiol Oncol. 2001; 35(3):185–91.
- Šajn L, Kononenko I, Milčinski M. Computerized segmentation and diagnostics of whole-body bone scintigrams. Computerized Medical Imaging and Graphics. 2007; 31(7):531–41. <u>https://doi.org/10. 1016/j.compmedimag.2007.06.004</u> PMID: 17683909
- Bishop CM, Hart PE, Stork DG. Pattern Recognition and Machine Learning. 1st ed. Berlin, Germany: Springer-Verlag New York; 2006.
- Does ImageDataGenerator add more images to my dataset? [Internet]. Available from: https://stackoverflow.com/questions/51748514/does-imagedatagenerator-add-more-images-to-my-dataset
- Albelwi S, Mahmood A. A framework for designing the architectures of deep Convolutional Neural Networks. Entropy. 2017;
- Ma L, Ma C, Liu Y, Wang X. Thyroid Diagnosis from SPECT Images Using Convolutional Neural Network with Optimization. Yasmin M, editor. Computational Intelligence and Neuroscience. 2019 Jan 15; 2019:6212759. https://doi.org/10.1155/2019/6212759 PMID: 30766599

- 74. Colaboratory cloud environment supported by Google [Internet]. Google Colab. Available from: https:// colab.research.google.com/
- 75. Chollet F. Xception: Deep Learning with Depthwise Separable Convolutions. 2016.
- 76. Howard AG, Zhu M, Chen B, Kalenichenko D, Wang W, Weyand T, et al. MobileNets: Efficient Convolutional Neural Networks for Mobile Vision Applications. 2017.
- 77. Russakovsky O, Deng J, Su H, Krause J, Satheesh S, Ma S, et al. ImageNet Large Scale Visual Recognition Challenge. International Journal of Computer Vision. 2015;
- 78. Deng J, Dong W, Socher R, Li L-J, Kai Li, Li Fei-Fei. ImageNet: A large-scale hierarchical image database. In 2010.